Author's response to reviews

Title: Expression of the costimulatory molecule B7-H3 is associated with prolonged survival in human pancreatic cancer

Authors:

Martin Loos (loos@chir.med.tu-muenchen.de)
Dennis M Hedderich (dennis.hedderich@gmx.de)
Malte Ottenhausen (MalteOttenhausen@web.de)
Nathalia N Giese (Nathalia.Giese@chirurgie.med.uni-heidelberg.de)
Melanie Laschinger (Laschinger@chir.med.tu-muenchen.de)
Irene Esposito (Irene.Esposito@helmholtz.de)
Jörg Kleeff (kleeff@chir.med.tu-muenchen.de)
Helmut Friess (Helmut.Friess@chir.med.tu-muenchen.de)

Version: 4 Date: 2 December 2009

Author's response to reviews:

Dear Sir or Madam,

As you can see on page 11/line 23, we already cited Table 1 at the relevant position in our manuscript.

We changed the authors' contributions according to your request, using only initials.

Thank you very much - warm regards from Munich,

Martin Loos